Author's response to reviews

Title: Urgent care centers in the U.S.: Findings from a national survey

Authors:

Robin M. Weinick (rweinick@partners.org)
Steffanie J. Bristol (sbristol@partners.org)
Catherine M. DesRoches (cdesroches@partners.org)

Version: 5 Date: 13 April 2009

Author's response to reviews:

Minor typographical errors have been corrected, including references remaining in a table, a table having footnote references using letters rather than asterisks, and one reference number being bolded. Many thanks.